Author's response to reviews

Title: Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation

Authors:

Serdar Demirgan (serdardemirgan@hotmail.com)
Kerem Erkalp (keremerkalp@hotmail.com)
M. Salih Sevdi (salihseydi@yahoo.com)
Meltem Turkay Aydogmus (meltem72_3@hotmail.com)
Numan Kutbay (nkutbay@hotmail.com)
Aydin Firincioglu (aydinfirincioglu@gmail.com)
Ali Ozalp (ali_ozalp3@hotmail.com)
Aysin Alagol (aysinalagol@yahoo.com)

Version: 4 Date: 26 May 2014

Author's response to reviews:

All persons listed as authors have contributed to preparing the manuscript, and no person or persons other than the authors listed have contributed significantly to its preparation.

The manuscript has been read and approved by all the authors and the requirements for authorship as stated earlier in this document have been met and each author believes that the manuscript represents honest work.

The manuscript has not been published elsewhere and has not been submitted simultaneously to another journal.

Kerem ERKALP, M.D.
On behalf of the authors.